Notes from RICH1 phone meeting 30/03/07

People present:

Bristol: Nick (part).

CERN: Carmelo, Dave, Kenneth, Olav, Richard.

Imperial: Bill, Gabriel.

Oxford: Rohan.

Apologies: Fabio, Neville.

1. Gas Enclosure.

No changes since the last meeting: previous outstanding items are still outstanding and are to be undertaken once more urgent items have been completed.  The door removal mechanism is now complete, and will be delivered to CERN next week (that starting Monday the 2nd April).

2.  Spherical Mirrors.

Mechanics:  The changes to the top rail are finished and the design complete.  The only outstanding item is the stops.  

The carbon fibre frame is now assembled. It differs from the drawing by about 1-2mm in the overall dimensions and care will be needed in setting up in the G.E. to ensure the correct clearances.  Measurements of feet position with respect to the rail have been made.  The feet need to be machined to give proper fit to rail.  All four are to be returned to Imperial for this.

3. PPSPS.

The documents for the mirror installation are done, with a need for minor changes.  Drafts for the HPDs are done, but the FEA calculations of the cantilevered frame are still outstanding.  These calculations should be received before Easter (action Rohan).

4.  Flat mirror rails and mounts.

The 2nd plate was assembled 29/03/2007, the PTFE bolts are not yet ready. This to be completed Monday 2nd April.

We do not have all the pieces for the rail mounts for the boxes.  There are doubts as to whether this will be done in time for the the shipment next week.

The production for plates and boxes is now almost complete, and will be completed by the start of the week commencing Monday 2nd April.

After Easter, most of the mechanics for the installation/trial installation/alignment should have arrived in CERN.  The last item is the angle adjuster for the flat mirror, which has gone out to manufacture at AC Precision.

5.  Optical alignment.

The frame for aligning both the flat and spherical mirrors has been assembled in building 156, although it still requires rails.

Test of the CF C-frame 23rd April or earlier, to be done before beam pipe commissioning.  Earlier is preferred.  A C-frame installation could be tried in building 156.  The slot for pit installation is Easter to the 29th April.  The 27th April to ~11th May is scheduled for commissioning of the beam pipe; alignment in B156 should be done then.

Some manufacturing may continue at Imperial after Easter - this is the critical item for trial assembly.

Trial installation for flat mirrors can take place 16th-20th April.  Tim  Newbury can be at CERN 16th-20th April and probably the end of May.

Richard can be at CERN from 7th to 11th May for alignment. As soon as boxes arrive they can be fitted.  As soon as rails arrive they can be fitted to the alignment rig.  A decision on whether to first set up for flat or spherical mirror alignment can then be made.

6.  HPDs.

Seven RICH1 columns have been completed and tested. Of the second batch, there are five RICH1 ladders which have been partially dressed.  Fibres are being installed.  After Easter all ladder parts will have been delivered to CERN.

Mechanics:  The workshop at Imperial has drawings for the chariot.  Machining has started.  Detailed drawings for the photon funnel will be sent out for quotation before Easter.  There is still design work for the patch panels on the end of the hood.  The sequence of work as fed into the workshop matches that of the assembly at CERN.  All anodising will be done before transporting.

Manufacture of rail system.  One quote has been received, but other manufacturers contacted have not accepted. The quote does not include the shelf rails and these need to be ordered.  (Action, Tony, Rohan, to pursue orders).
Cabling:  Vera and Maria are coming to the end of their two-week working visit at CERN.  Connectors have been put on the RICH1 cables.  There has been a problem with the laying of the cables due to the missing upper cable tray.  It is possible that this cannot be fixed before the middle of May.

7.  MDMS.

The results of the last meeting where as follows:  There has been progress towards compatibility of moving stage and the photon funnel, but further work is still needed on this front.  Flexibility is gained from having the gear box rotated by 90 degrees w.r.t. the motor, and this may allow for resolution of clashes with photon funnel.  The manufacturer is able to extend the length of the movable stage.  There is another meeting on Tuesday (03/04/07).

8. LAMS.

Focalisers and splitters to be ordered next week (beginning 02/04/07).  The fibres have been installed from the barrack but need to be tested for damage.  The fibres currently going to wrong rack, and need to be reshuffled.

9. Aerogel.

A trial installation of the boxes is planned for week 16-20 April.
